Iowa State University 1 CprE 489, Lecture 23

CprE 489 Lecture 23

Fall Semester 1997

Iowa State University 2 CprE 489, Lecture 23

Transport Flow Control

• For reliable transmission

- Use a sliding window like protocol with selective repeat. - With a byte stream protocol an acknowledgement does not necessarily mean that the sender should send more data. The received data may not have been consumed by the application yet.

• For timely transmission

Assume the sender transmits at a relatively steady rate of x bytes per second. We want the receiver to deliver the data at the same rate with some delay, but with possible gaps. receiver’s buffer space is a window sliding at x bytes per second

Iowa State University 3 CprE 489, Lecture 23

TCP Congestion Control

• Fast network feeding to low capacity receiver • Slow network feeding a high capacity receiver • Threshold with time out management

Iowa State University 4 CprE 489, Lecture 23

Timeouts

• Retransmission timer - when to resend if ack has not arrived - must be variable not fixed because round-trip delay can vary - RTT = a RTT + (1-a) M commonly used value is a = 7/8. RTT is the current estimate of round trip delay. M is the most recent round trip delay. - timeout period could be say 2 RTT or perhaps RTT + 4 D where D is an estimate of the current mean variation of the delay.

• Persistence timer - a packet to the sender may tell it to stop transmitting.

The packet from the receiver telling it to restart again might get lost so it needs a timer.

• Keepalive timer - when a connection has been idle for a long time a packet may be needed to check to see if the other party is still there.